Author's response to reviews

Title: Association between TP53 codon 72 polymorphism and risk of oral squamous cell carcinoma in Asians: A meta-analysis

Authors:

- Xian-Tao MM Zeng (zengxiantao1128@163.com)
- Wei MD Luo (286806931@qq.com)
- Pei-Liang MD Geng (1057179657@qq.com)
- Yi MD Guo (guoyi@whu.edu.cn)
- Yu-Ming MD Niu (n4oneone@126.com)
- Wei-Dong MD Leng (lengtaihe@163.com)

Version: 5 Date: 10 June 2014

Author's response to reviews: see over
Dear Ms Roselyn Remoto and The BMC Cancer Editorial Team,

We thank you sincerely for highlighting the further grammatical errors that remained in our resubmission. We have indeed recruited a voluntary native English expert with sound experience in scientific publication who has revised the manuscript extensively; the revisions are highlighted in red. We apologize for the oversights (e.g. re. the PRISMA statement) and we hope that we have shown great commitment in publishing our article in your prestigious journal.

If you insist that we should use Edanz (www.edanzediting.com/bmc1) to edit our manuscript, please tell us and we will ask Edanz for help.

Once again, thank you very much for your comments and suggestions.

Best Regards.

Yours Sincerely,

Wei-Dong Leng

E-mail: lengtaihe@163.com